Author's response to reviews

Title: Feasibility and Accuracy Evaluation of Three Human Papillomavirus Assays for FTA Card-based Sampling: a Pilot Study in Cervical Cancer Screening

Authors:

Shao-Ming Wang (wangshaoming@cicams.ac.cn)
Shang-Ying Hu (shangyinghu@cicams.ac.cn)
Wen Chen (chenwen@cicams.ac.cn)
Feng Chen (chenfeng915@tom.com)
Fang-Hui Zhao (zhaofangh@cicams.ac.cn)
Wei He (545875763@qq.com)
Xin-Ming Ma (maxinming2009@163.com)
Yu-Qing Zhang (yqzhang2010@hotmail.com)
Jian Wang (591430941@qq.com)
Priya Sivasubramaniam (priyagsiva@gmail.com)
You-Lin Qiao (qiaoy@cicams.ac.cn)

Version: 6 Date: 19 October 2015

Author's response to reviews:

Dear Editor,

Thank you very much for your letter and the accompanying comments on our paper (Manuscript ID 8196866791587546). We are submitting a revised manuscript that incorporates the comments. A point-by-point response is attached below.

Comment:

Please remove the EC approval document from the supplementary files.

Response:

We have deleted the EC approval letter from the supplementary files.

We hope that our paper has been revised satisfactorily and will be published in BMC Cancer.

We look forward to your response.

Sincerely,

You-Lin Qiao M.D. Ph.D.
Shao-Ming Wang M.D. Ph.D.